Author's response to reviews

Title: Associations of food group and nutrient intake, diet quality, and meal sizes between adults and children in the same household: a cross-sectional analysis of U.S. households

Authors:

Jennifer L Zuercher (jzuercher2007@gmail.com)
David A Wagstaff (daw22@psu.edu)
Sibylle Kranz (kranz@purdue.edu)

Version: 5 Date: 23 November 2011

Author's response to reviews: see over
November 22, 2011

Nehme Gabriel
Editor-in-Chief
Nutrition Journal

Attached please find our manuscript "Associations of food group and nutrient intake, diet quality, and meal sizes between adults and children in the same household: a cross-sectional analysis of U.S. households," co-authored by Jennifer Zuercher, David Wagstaff, and Sibylle Kranz. We have addressed the additional major and minor editorial revisions as requested. We hope that we have addressed all of your comments and concerns with this version of the manuscript. Thank you.

Sincerely,

Sibylle Kranz, PhD RD (Corresponding author)
Associate Professor, Member of the Ingestive Behavior Research Center
Department of Foods and Nutrition Purdue University
204 Stone Hall, 700 W. State Street
West Lafayette, IN 47907
phone: 765-494-6758
email: Kranz@purdue.edu
Formatting changes to be made:

-----------------------------

Major revisions (we require the author to make these changes)

Tables: The first table referenced in the manuscript text should be labelled Table 1. The second, Table 2, etc. The tables should be provided in this same order at the end of the manuscript file. Every table provided at the end of the manuscript file must be referred to in the text. Please do not use ranges when listing tables. Please list all the figures, for example: Tables 5, 6, 7, 8 and 9. Currently, table 6 is not cited in the main text. Please ensure that this table is cited in numerical order.

Response: We apologize for not making this correction accurately in the previous version of the manuscript. We believe that we have now corrected the issue.

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

Figures: It is important for the final layout of the manuscript that the figures are cropped as closely as possible to minimise white space around the image. Our online figure guide contains full details for preparing files for submission and can be viewed here: http://www.biomedcentral.com/info/ifora/figures.

Response: We have further reduced the white space around the figure. We hope that we have cropped the figure satisfactorily.

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.

Response: Thank you for the opportunity to review our manuscript once again. We believe that all typographical errors have been addressed and no further corrections or revisions are necessary.